Author's response to reviews

Title: Efficacy of vitamin D3-fortified-yogurt drink on anthropometric, metabolic, inflammatory and oxidative stress biomarkers according to vitamin D receptor gene polymorphisms in type 2 diabetic patients: a study protocol for a randomized controlled clinical trial

Authors:

Sakineh Shab-bidar (s_shabbidar@razi.tums.ac.ir)
Tirang R Neyestani (neytr@yahoo.com)
Abolghassem Djazayery (jazaiers@sina.tums.ac.ir)

Version: 2 Date: 14 June 2011

Author's response to reviews:

To whom it may concern:

All the corrections were made. References 45-7 had been addressed in table 2. However, they were cited in the maintext too. I also added 5 key words in the end of the abstract.